Author's response to reviews

Title: An online tool for evaluating diagnostic and prognostic gene expression biomarkers in bladder cancer

Authors:

Garrett M Dancik (dancikg@easternct.edu)

Version: 2 Date: 30 January 2015

Author's response to reviews:

I have made the following revisions in order to meet the required format prior to editorial assessment.

1. Line numbers were added.
2. The location of Table 1 was moved from before to after the figure legends.
3. A Competing Interests section was added.
4. A statement regarding human subjects and Institutional Review was added to the Construction and Content Section.